Author's response to reviews

Title: Pediatric appendicitis rupture rate: a national indicator of disparities in healthcare access

Authors:

Kathleen A Jablonski (kjablons@gwu.edu)
Mark F Guagliardo (markg@gwu.edu)

Version: 2 Date: 2 May 2005

Author's response to reviews:

Editorial formatting revisions only